Author's response to reviews

Title: Dose patterns in commercially insured subjects chronically exposed to opioids: a large cohort study in the United States

Authors:

Maria Soledad Cepeda (scepeda@its.jnj.com)
Mila Etropolski (metropol@its.jnj.com)
Rachel Weinstein (rweinst1@its.jnj.com)
Daniel Fife (dfife@its.jnj.com)
Raymond Boston (drrayboston@yahoo.com)
Amy Matcho (amatcho@its.jnj.com)

Version: 12 Date: 17 June 2010

Author's response to reviews: see over
I made two changes, changed a capital letter and added a space between subheadings